Author's response to reviews

Title: Efficacy and tolerability of PEG-only laxative on faecal impaction and chronic constipation in children: a randomized controlled study vs a standard PEG-electrolyte laxative.

Authors:

Francesco Savino (francesco.savino@unito.it)
Serena Viola (viola.serena@libero.it)
Erasmo Miullari (erasmo.maiullari@oirmsantanna.piemonte.it)
Giovanni Di Nardo (giovanni.dinardo@uniroma1.it)
Salvatore Oliva (salvatore.oliva@uniroma1.it)
Salvatore Cucchiara (salvatore.cucchiara@uniroma1.it)

Version: 9 Date: 2 November 2012

Author's response to reviews: see over
Letter for the reviewer

Efficacy and tolerability of PEG-only laxative on faecal impaction and chronic constipation in children: a randomized controlled study vs a standard PEG-electrolyte laxative MS: 9353636207284261

We thank the reviewers for their thoughtful suggestions. Please find below an itemized response to all the reviewer comments:

Response to Reviewer 2 (Mike Geraint)

1) Comment on the abstract
The text has been changed according the request of the reviewer 2.

We also change the phrase of the abstract.

“At the higher PEG doses recommended by the manufactures, children in the PEG-only group had higher and more regular soft stool frequency than PEG-EL.” Page 3.

2) Page 9 last paragraph
The sentence is as requested by the reviewer 2.